Research Paper



Assessment and Treatment of Childhood Apraxia of Speech: An Inquiry into Knowledge and Experience of Speech-Language Pathologists

Mersede Imani-Shakibayi 1 (0), Talieh Zarifian1* (0), Nasibeh Zanjari2 (0)

1. Department of Speech Therapy, University of Social Welfare and Rehabilitation Sciences, Tehran, Iran.

2. Iranian Research Center of Aging, University of Social Welfare and Rehabilitation Sciences, Tehran, Iran.



Citation Imani-Shakibayi M, Zarifian T, Zanjari N. Assessment and Treatment of Childhood Apraxia of Speech: An Inquiry into Knowledge and Experience of Speech-Language Pathologists Iranian Rehabilitation Journal. 2022; 20(1):1-10. http:// dx.doi.org/10.32598/irj.20.1.24.4

doi http://dx.doi.org/10.32598/irj.20.1.24.4

\odot \odot

Article info: Received: 12 May 2020 Accepted: 06 Jan 2021 Available Online: 01 Jun 2022

Keywords:

Childhood apraxia of speech, Assessment, Treatment, Prognosis, Speech-language pathologists

ABSTRACT

Objectives: The present research aims to identify the assessment and treatment processes used by Iranian Speech-Language Pathologists (SLPs) for Childhood Apraxia of Speech (CAS) and investigate the impact of their knowledge level and experience on their choice of assessment and treatment.

Methods: This research is a cross-sectional study using a survey design conducted on 260 SLPs with a minimum of a Bachelor's degree and at least one year of experience of working with preschoolers. The CAS assessment and treatment were measured by a validated questionnaire, which was completed in person or online.

Results: The tests of Diadochokinesis (DDK) (66%), single-word speech sampling (58.1%), oral-motor assessment (54.6%), and connected speech sample analysis (53.1%) were the popular tests chosen by the participants. The treatment approaches indicated that Oral Motor Exercises (OMEs) (57.7%) were the only treatment for which over half of the participants voted. The experts chose phonologically-based treatments and Integrated Phonological Awareness (IPA), but the less-experienced participants were more interested in PROMPT (prompts for restructuring oral muscular phonetic targets). The majority of the participants (70.8%) believed that children with CAS make very slow progress and 21.9% declared that speech problems of such children persist through the school years.

Discussion: The participants' choice of assessment tasks is in line with the results of recent studies. However, opting for outdated treatments such as OME indicates a gap between the clinicians' knowledge and experience in using evidence-based treatments.

* Corresponding Author: Talieh Zarifian, PhD. Address: Department of Speech Therapy, University of Social Welfare and Rehabilitation Sciences, Tehran, Iran. Tel: +98 (21) 22180043 E-mail: t.zarifian@yahoo.com ; ta.zarifian@uswr.ac.ir

Highlights

• Consistent with international studies, the Iranian Speech-Language Pathologists (SLPs) prioritized Diadochokinesis (DDK), single-word speech sampling, oral motor assessment, and connected speech analysis in diagnosing Childhood Apraxia of Speech (CAS).

• In the present inquiry, most participants prioritized Oral Motor Exercises (OMEs) as the main intervention approach for treating CAS; however, there is no convincing evidence to support its use.

Plain Language Summary

Children with CAS are far from being a homogenous population. This study surveyed the SLPs' clinical experience in evaluating and intervening in the speech problems of this group of children. The present study results showed that although there is no clear guideline for identifying actual cases of CAS, Iranian clinicians follow the correct route aligned with the scientific findings. Research findings suggest that insufficient university training or professional development courses may use outdated treatments in managing CAS.

1. Introduction

hildhood Apraxia of Speech (CAS) is a neurological speech sound disorder targeting the accuracy and consistency of speech movements without notifying any neuromuscular defects [1]. Because of the overlap between the speech characteris-

tics of this disorder and other Speech Sound Disorders (SSDs), Speech-Language Pathology (SLP) faces numerous challenges in the differential diagnosis. The absence of a standard evaluation process, which harbors a host of clinical decisions such as diagnosis, determining the intensity, prognosis, and treatment planning, is the major problem in the differential diagnosis of CAS [2].

An overview of different evaluation methods in CAS

Through a systematic review, Gubiani, Pagliarin, and Keske-Soares (2015) [3] explored the available diagnostic tests for CAS. Their results indicated that a few tests had been designed for CAS, and there are very few international studies on the subject. None of the formal evaluation tools have psychometric specifications to evaluate CAS except for Dynamic Evaluation for Motor Speech Skills (DEMSS) [3]. The previous SLP surveys have revealed that in countries with languages other than English, informal or self-made assessment tools are commonly used. Proofs of this finding could be maximal administration of informal assessments by South African SLPs [4], US SLPs [5], and Swedish SLPs [6]. Murray et al. suggested the polysyllabic word test and oral-motor examination containing Diadochokinesis (DDK) to identify CAS with 91% of confidence [7].

Regarding CAS evaluation methods, SLP surveys are restricted to only two studies. Dawson considered the assessment and treatment for CAS in children with autism by 132 American SLPs and noted that at least half of the participants implemented the assessments of connected speech, oral mechanism test, DDK, and imitation of polysyllabic words. They also tended to use informal scales, probably due to the difficulty of performing formal evaluations for children with autism [5]. A Swedish survey, on the other hand, reported a variety of choices on assessment methods indicating an incongruence of opinions among Swedish clinicians. Nearly half of the Swedish SLPs used formal perceptual analysis plus oromotor assessment, and the rest chose their assessment based on their perception of what the origin of CAS might have been, namely, motor-based or languagebased. Moreover, only half of them had diagnosed and assessed CAS, among which nearly two-thirds were confident about their approach [6].

An overview of treatment approaches for CAS

Since the American Speech-Language-Hearing Association (ASHA) technical report of 2007 somehow settled a part of disagreements over CAS, instances of CAS studies have risen drastically, and different treatments have been proposed for the disorder. Review articles have categorized treatment approaches into three groups of motor treatments, linguistic approaches, and Augmentative and Alternative Communication (AAC) [8]. The most outstanding evidence-based treatment approaches are summarized in Table 1.

Therapy Type	Intervention Approach	Cases With Reported Effect	Measures	Judgment of Certainty
Motor-Based	Integral stimulation/DTTC Combined MIT with TCM ReST	11/13 1/1 3/3	Accuracy PVC/PCC/PMLU/PWP/PWC Perceptual stress marches	Preponderant Suggestive Preponderant
Linguistic- Based	IPA Combines STP with mCVT	11/15 4/4	Suppression of process usage PA accuracy PCC Phones added to inventory Inconsistency decrease	Preponderant Suggestive
AAC	Modeling with board Computer-based AAC	4/4 1/1	Morpheme accuracy Multisymbol messages Accuracy in: Book reading Discourse	Suggestive Suggestive

Table 1. Varying treatment approaches to CAS and the research results on each

ranian Rehabilitation Journal

DTTC: Dynamic Tactile and Temporal Cueing; MIT: Melodic Intonation Therapy; TCM: Tactile Cueing Method; ReST: Rapid Syllable Transition; IPA: Integrated Phonological Awareness; mCVT: modified Core Vocabulary Treatment; PVC: Percentage of Vowels Correct; PCC: Percentage of Consonant Correct; PMLU: Phonological Mean Length of Utterance; PWP: Proportion of Whole-Word Proximity; PWC: Percentage of Word Correct; PA: Phonological Awareness.

Regarding CAS treatment, SLP surveys are limited to just two choices. AAC and PROMPT (prompts for restructuring oral muscular phonetic targets) were the two treatment approaches used more extensively by US clinicians for children with autism suspecting CAS. The study findings indicated that few participants used integral stimulation, and nearly one-third of them implemented Oral Motor Exercises (OMEs) [5]. In a more recent survey on CAS treatment, Gomez, McCabe, and Purcell reflected the preference of SLPs from Australia and New Zealand for eclectic treatments, while the treatment efficacy in CAS cases has not been proved empirically. It is worth mentioning that the most common treatment in Australia is reported to be a Nuffield Dyspraxia Program [9].

Prognosis of CAS

Because clear results of CAS prognosis are not handy in the literature, the oral reports of therapists on the considerable challenge of treatment have been used as the last resort. In a Swedish survey, the majority of the respondents believed that children with CAS improve rather slowly. The majority of participants voted for the constancy of speech results at school age; more than half of them, as well, reported the children resisted treatment and had a backslide of the symptoms after a remission period [6].

Speech-language pathology faces immense challenges in the valid differential diagnosis between CAS and other Speech Sound Disorders (SSDs). Major reasons could be the absence of a standard evaluation process and distinct inclusion criteria to identify such children. The body of research in Iran is desperate for studies on the identification, formal assessment, and treatment of CAS. Since using the clinical experience of SLPs in clinical decision-making is of sublime diagnostic and treatment values, investigating the assessment and treatment approaches of children with CAS in the absence of formal assessments in Persian seems mandatory. Thus, investigating Iranian SLPs' performance regarding the evaluation and treatment of CAS was the purpose of this research. It was expected that the viewpoints of the practicing SLPs participated in the present survey, along with international study results, would set the grounds for helping researchers design appropriate assessment packages in Persian.

Consequently, the present study intends to answer the following questions: What are the assessment and treatment approaches used by Iranian SLPs regarding CAS? What is the relationship between their knowledge and years of experience? For what assessment and treatment approach do they opt? How confident are they about the assessment and treatment approaches they choose? And what prognosis do the participants assume for children with apraxia?

2. Materials and Methods

Study participants

This cross-sectional study implemented a researchermade questionnaire to access the knowledge and experience of Iranian SLPs regarding the assessment and treatment of CAS. The inclusion criteria for participating in the study were holding a minimum Bachelor's degree in speech-language pathology and at least one year of experience working with preschoolers. The questionnaires were excluded if they were filled out incompletely, not completed by a practicing SLP, and if the participants were not treating preschool children.

Questionnaire development

The initial draft of the questionnaire was developed via a literature review [4, 6, 10]. A group of 10 SLP experts in the field of CAS evaluated the face and content validity of items. The Content Validity Ratio (CVR) was found to be acceptable (>0.62) for most items except 3 questions which were rated as "not necessary" by more than half of the experts and thus were omitted from the instrument. The final version of the questionnaire included 18 items with different questions: yes/no, Likert-type scale, short answer, and multiple-choice questions. Questions 1 to 6 asked about the demographic information, including educational background, years of clinical experience, place of work, the work setting, and passing or not the CAS course. Questions 7 to 9 targeted the amount of CAS referrals and caseloads of the participants. The clinicians' knowledge of the definition and etiology of the CAS was assessed through two multiple-choice questions based on definitions provided by the ASHA (Qs 10-11). The four multiple-choice questions check the participants' knowledge and experience regarding CAS disorder included choosing speech characteristics of CAS (from the 14 speech features from the literature) (Q 12), choosing the co-occurring problems (Q 13), choosing common test(s) appropriate for CAS from 10 common SSD tests (Q 14), choosing a treatment approach or approaches currently in practice (from eight treatment approaches) (Q 15), and choosing prognosis of children with CAS (from the provided four items) (Q 16). The participants were also asked to rate their ability to diagnose and treat the disorder based on a 5-point Likert scale ranging from the least confident to the most confident (Qs 17-18). The questions regarding assessment, treatment, and prognosis of the CAS are considered in this article.

Study procedure

The questionnaires were distributed among SLPs either in print at the workshops and conferences or online (hosted on http://www.cafepardazesh.ir). They were accompanied by a letter explaining the study subject and objectives. SLP participants were invited via email or social media posts by the Iranian Scientific Association of Speech-Language Pathology among practicing members, allowing a 3-month response time. A total of 260 questionnaires, according to the inclusion criteria, were rendered valid for analysis.

Statistical analysis

Data analysis, including descriptive and inferential statisfics, was conducted using SPSS software v. 20.0. Descriptive statisfics (frequency and percentage) were used to illustrate the characterisfics of the participants and their knowledge of CAS assessment and treatment. The inferential statistics (bivariate analysis, the Chi-square) were performed to investigate any possible significant relationships between the study variables. The level of statistical significance was set at P<0.05.

3. Results

The results of the addressing questions of the study are presented under the three subtitles of assessment methods, treatment approaches, and prognosis.

Study participants

The participants were 179 Bachelor's holders, 61 Master's holders, and 20 PhD holders in speech therapy. They were divided into three groups: less-experienced (one to three years of experience, n=107), experienced (four to ten years of experience, n=95), and expert (more than ten years of experience, n=58) [10]. The participants came from different clinical settings, including hospitals (14.6%), general or daily rehabilitation centers (17.3%), public speech therapy clinics (8.7%), private speech therapy clinics (62.3%), exceptional schools (20.4%), and other places (8.8%). The participants' characteristics are summarized in Table 2.

Results of assessment methods

Of the ten considered tests used to diagnose SSDs, eight demonstrated sufficient reliability and validity in Persian. The frequency of each assessment voted by the participants is presented in Figure 1. DDK test was the main CAS assessment task with 66% of the votes, and the articulation test [11] displayed the least favorability, with 18.8% voting. Selecting the type of assessment task was not significantly related to educational level, but the relationship between work experience and using DDK and inconsistency test [12] proved to be significant. Both DDK (χ^2 =8.840, P<0.01) and inconsistency (χ^2 =9.062, P<0.01) tests were selected mainly by the less experienced SLPs. However, those confident participants

Table 2. Characteristics of respondents (N=260)

Charac	Frequency of Respondents (%)	
	Bachelor	68.8
Level of education	Master	23.5
	PhD	7.7
	1-3 years (less-experienced)	41.2
Work experience	4-10 years (experienced)	36.5
	Above 10 years (expert)	22.3
	Hospital	5.4
	University hospital	9.2
	Rehabilitation center	17.3
Work setting	Public speech therapy clinic	8.8
	Private speech therapy clinic	62.3
	Exceptional school	20.4
	Other working places	8.8
Diagnocod childron with CAS	Yes	87.7
Diagnosed children with CAS	No	12.3

about their diagnosis tended to select the articulation test (χ^2 =19.881, P<0.001).

Results of treatment approaches

Among the eight approaches presented as the common SSD and CAS treatments, OMEs (57.7%) were the most selected, and Rapid Syllable Transition (ReST) (18.8%) was the least favored. Table 3 presents SLPs' frequency of use of treatment approaches. Regarding the treatment preference, only traditional articulation Iranian Rehabilitation Journal

therapy [13] was negatively correlated with educational level. Compared to others, participants with a BA degree tended to go for traditional articulation therapy for CAS intervention (χ^2 =4.668, P<0.009). Investigations on the relationship between work experience and treatment approaches revealed that only phonologically based approaches, PROMPT, and Integrated Phonological Awareness (IPA) significantly correlated with work experience. The experts placed a high priority on selecting phonologically-based approaches (χ^2 =13.827, P<0.001) and IPA (χ^2 =28.254, P<0.0001), whereas the

Table 3. Frequency of votes on each CAS treatment approach as decided by the SLPs participating in the research (N=260)

Treatment Approaches	Frequency of Respondents (%)
Non-Speech Oral Motor Exercises (NSOMEs)	57.7
PROMPT	46.5
Core Vocabulary Treatment (CVT)	41.2
Traditional articulation therapy	35
Dynamic Tactile and Temporal Cueing (DTTC)	28.1
Integrated Phonological Awareness (IPA)	28.1
Phonologically based treatments	21.9
Rapid Syllable Transition (ReST)	18.8

Iranian Rehabilitation Journal



Figure 1. Frequency of votes on each test as decided by the survey's participants (n=260)

DDK: Diadochokinesis; Single-word SS: Single-word Speech Sampling; CS: Connected Speech; NRT: Nonwords Repetition Test; Polysyllab: Polysyllabic word test; SRT: Syllable Repetition Test; Inc: Inconsistency test of DEAP; Phono: phonological test of DEAP

less-experienced ones had a significant inclination towards PROMPT (χ^2 =21.148, P<0.0001). Interestingly, higher work experience coincided with an increase in selecting OMEs and traditional articulation therapy. The confident participants about their choice of treatment selected PROMPT, Core Vocabulary Treatment (CVT), and IPA in this order.

Results on prognosis

Participants were invited to choose one of the four options that were given to them for assessing their experiences with treatment results. Frequencies of the participants' choices regarding the prognosis of CAS disorder are presented in Figure 2. The experts had no hopes for CAS children to be treated (χ^2 =5.271, P<0.0072), and those who felt confident about choosing an appropriate treatment were more certain that such children would never be cured (χ^2 =35.674, P<0.0001). Although a significant relationship was not reported between educational level and CAS prognosis-related questionnaire items, the more educated participants selected the following statement "these children reach a complete improvement".

4. Discussion

This study aimed to consider the clinical decisions of Iranian SLPs regarding CAS assessment and treatment approaches and investigate the relationship between background knowledge and experience of participants and their assessments and treatments, and their stance on the disorder's prognosis.

Assessment methods

The question of assessment methods revealed that Iranian SLPs prioritized DDK, single-word speech sampling [14], oral motor, and connected speech analysis. Selecting the task of DDK as CAS's main evaluation tool by the Iranian clinicians, especially those who had recently graduated, is consistent with the findings of Lewis et al.'s and Murray et al.'s studies and the US survey [5, 7, 15].



Figure 2. Frequency of votes on each CAS prognosis as decided by the participants (N=260)

Iranian Rehabilitation Journal

The participants placed second priority on the singleword speech sampling test, indicating that they pay close attention to the necessity of eliciting children's production information. Additionally, due to its good recognition, availability, and easy application, this test is more widely favored in Iran than Goldman-Fristoe or the articulation test.

The favorability of oromotor assessment demonstrates that clinicians have always had in mind that the oral motor problems of CAS root in motor planning deficit. In the current study, those who were quite confident about their knowledge and experience tended to choose oral motor assessment. Such a choice has been confirmed by Lewis et al., Strand and McCauley, Iuzzini and Forrest, and Murray et al. They all emphasized the importance of oromotor assessment as one of the major diagnostic approaches of CAS [7, 15-17].

In this research, favoring the task of deriving the connected speech sample indicates that Iranian SLPs are drawn to a more real sample of children's production. It is still possible that the lack of standardized tests and their availability in Iran are driving clinicians toward broader use of connected speech analysis. The same justification can be applied to Dawson's conclusion regarding the inability of children with autism to take structured tests as the probable reason for choosing connected speech analysis to diagnose CAS [5]. Further, Australian SLPs have reported that they use detailed transcription more often when recording the speech of children with CAS [18].

As it is challenging, if not impossible, to obtain and transcribe an appropriate speech sample due to the unintelligibility of the child, a polysyllabic word test has been suggested to acquire a more realistic sample of speech and induce motor challenges for the production of speech [7, 19]. Such assessment has been referred to as one of the most fundamental evaluations of CAS by Murray et al. and Dawson [5, 7]. However, almost a third of Iranian SLPs voted for the test, possibly because of the absence of formal polysyllabic words test in Iran.

Although Iranian practitioners acknowledged inconsistency as the main characteristic of CAS [20], fewer SLPs prioritized the inconsistency test of DEAP for assessment. This choice is supported by recent studies by Luzzini and Forrest, Murray et al., and Luzzini et al., which have not endorsed the validity of the DEAP inconsistency test in the diagnosis of CAS [7, 17, 21].

Treatment approaches

An abundance of treatment approaches for CAS makes it very difficult to select the appropriate one, but it is known that the top priority of choosing an approach is to eliminate the fundamental deficits. In the present inquiry, most participants prioritized OMEs, although we know that orofacial structural impairments are not associated with the CAS diagnosis, and no convincing evidence shows this treatment improves the verbal skills in such children [22]. The data analysis traced a declining trend in implementing OMEs with an increase in educational level versus an ascending trend as years of experience increased. Dawson (2010) also analyzed practitioners' ideas in their respective studies and concluded similarly, which could demonstrate experts' lack of awareness about evidence-based treatments and their inclination toward traditional approaches. Although PROMPT was Iranian SLPs' second choice and preferred treatment of US SLPs, less than one-sixth of Australian therapists reported using it to treat CAS cases [5, 9]. The treatment has certain limitations that make it usable only to certified people through passing authorized educational courses. Therefore, an increase in selecting PROMPT by respondents is due to their information acquired through reading books and articles. Choosing CVT with a high frequency by the respondents is thought-provoking. This intervention has been introduced for treating an inconsistent phonological disorder in the absence of CAS [23], and there is no evidence of using it for the treatment of CAS in the literature. The two motor-based treatments of Dynamic Tactile Temporal Cueing (DTTC) and ReST, which are highly convincing in terms of evidential resolution and generalization rigor, were chosen the least by Iranian SLPs. Gomez et al. came up with similar results and reported that less than 15% of Australian SLPs voted for the treatments mentioned above [9].

Prognosis of CAS

A high percentage of Iranian SLPs reported children with CAS as slow to improve. The majority of SLPs in the Swedish survey voted for the same concept [6]. Around one-fourth of the participants believed that the disorder persists at school age. This is what the Swedish survey portrayed with higher vote frequencies. Studies that picture such children as prone to reading, educational, social, and occupational problems also affirm and support the attained results [24, 25].

5. Conclusion

Consistent with other countries' clinicians and recent studies, Iranian SLPs are more inclined to use informal tests, leading researchers to devise efficient and easy-toimplement assessment tools. Although it was expected that the clinicians would share their clinical experience about the efficient treatment of CAS, older treatments with low levels of evidence were reported. The reasons can be attributed to lack of information resources, lack of standardized assessments, and limited national and international CAS-related studies.

Ethical Considerations

Compliance with ethical guidelines

Human Research Ethics Committee of the University of Social Welfare and Rehabilitation Sciences approved this research (Code: IR.USWR.REC.1397.141).

Funding

This work was supported by the University of Social Welfare and Rehabilitation Sciences.

Authors' contributions

Conceptualization and Supervision: Talieh Zarifian; Methodology: Nasibeh Zanjari; Investigation, Writingoriginal draft: Mersede Imani Shakibayi; Writing-review &editing: all authors; Data collection: Mersede Imani Shakibayi; Data analysis: All authors.

Conflict of interest

The authors of the manuscript declared no conflict of interest. The authors are responsible for the content and writing of the paper.

Acknowledgments

We thank the SLP colleagues who participated and shared their clinical experiences. We would also like to thank the Iranian Scientific Association of Speech Therapy for distributing the questionnaire among its members.

References

 American Speech-Language-Hearing Association. Childhood apraxia of speech [Position staement] [Internet]. 2007. http://www.asha.org/policy/ps2007-00277.htm

- [2] Iuzzini-Seigel J, Murray E. Speech assessment in children with childhood apraxia of speech. Perspectives of the ASHA Special Interest Groups. 2017; 2(Part 2):47-60. [DOI:10.1044/ persp2.SIG2.47]
- [3] Strand EA, McCauley RJ, Weigand SD, Stoeckel RE, Baas BS. A motor speech assessment for children with severe speech disorders: Reliability and validity evidence. Journal of Speech, Language, and Hearing Research: JSLHR. 2013; 56(2):505-20. [DOI:10.1044/1092-4388(2012/12-0094)]
- [4] Pascoe M, Maphalala Z, Ebrahim A, Hime D, Mdladla B, Mohamed N, et al. Children with speech difficulties: An exploratory survey of clinical practice in the Western Cape. The South African Journal of Communication Disorders. 2010; 57:66-75. [DOI:10.4102/sajcd.v57i1.51] [PMID]
- [5] Dawson EJ. Current assessment and treatment practices for children with Autism and suspected childhood apraxia of speech: A survey of speech-language pathologists [MSc. Thesis]. Portland: Portland State University; 2010. [DOI:10.15760/etd.29]
- [6] Malmenholt A, Lohmander A, McAllister A. Childhood apraxia of speech: A survey of praxis and typical speech characteristics. Logopedics, Phoniatrics, Vocology. 2017; 42(2):84-92. [DOI:10.1080/14015439.2016.1185147] [PMID]
- [7] Murray E, McCabe P, Heard R, Ballard KJ. Differential diagnosis of children with suspected childhood apraxia of speech. Journal of Speech, Language, and Hearing Research: JSLHR. 2015; 58(1):43-60. [DOI:10.1044/2014_JSLHR-S-12-0358] [PMID]
- [8] Murray E, McCabe P, Ballard KJ. A systematic review of treatment outcomes for children with childhood apraxia of speech. American Journal of Speech-Language Pathology. 2014; 23(3):486-504. [DOI:10.1044/2014_AJSLP-13-0035] [PMID]
- [9] Gomez M, McCabe P, Purcell A. Clinical management of childhood apraxia of speech: A survey of speech-language pathologists in Australia and New Zealand. International Journal of Speech-Language Pathology. 2019; 21(3):295-304. [DOI:10.1080/17549507.2019.1608301] [PMID]
- [10] Joffe V, Pring T. Children with phonological problems: A survey of clinical practice. International Journal of Language & Communication Disorders. 2008; 43(2):154-64. [DOI:10.1080/13682820701660259] [PMID]
- [11] Zarifian T, Modarresi Y, Gholami Tehrani L, Dastjerdi Kazemi M, Salavati M, Sadeghi A, et al. Persian articulation assessment for children aged 3-6 years: A validation study. Iranian Journal of Pediatrics. 2017; 27(4):e8217. [DOI:10.5812/ijp.8217]
- [12] Zarifian T. [Adaptation of diagnostic evaluation of articulation and phonology for 3-6 Persian speaking children and determining it's psychometric properties (Persian)]. Tehran: University of Social Welfare and Rehabilitation; 2014. https://ganj.irandoc.ac.ir/viewer/077b9d6761b11810350aaa2fe 9b5e14c?sample=1
- [13] Van Riper C. Speech correction principles and methods. 2th ed. New York: Prentice Hall; 1939. https://books.google. com/books/about/Speech_Correction_Principles_and_ Methods.html?id=_904AAAAMAAJ
- [14] Ghasisin L, Ahmadi T, Mostajeran F, Moazam M, Derakhshande F. [Evaluating the reliability and validity of phonetic information test in normal 3-6 year-old children of Isfahan city (Persian)]. Journal of Research in Rehabilitation Sciences. 2013; 9(2):153-60. https://www.sid.ir/en/journal/ViewPaper.aspx?id=334118

- [15] Lewis BA, Freebairn LA, Hansen AJ, Iyengar SK, Taylor HG. School-age follow-up of children with childhood apraxia of speech. Language, Speech, and Hearing Services in Schools. 2004; 35(2):122-40. [DOI:10.1044/0161-1461(2004/014)]
- [16] Strand EA, McCauley RJ. Differential diagnosis of severe speech impairment in young children. The ASHA Leader. 2008; 13(10):10-3. [DOI:10.1044/leader.FTR1.13102008.10]
- [17] Iuzzini J, Forrest K. Evaluation of a combined treatment approach for childhood apraxia of speech. Clinical Linguistics & Phonetics. 2010; 24(4-5):335-45. [DOI:10.3109/02699200903581083] [PMID]
- [18] Nelson Taylor L, Mok Z, Ttofari Eecen K. Use of transcription when assessing children's speech: Australian speech-language pathologists' practices, challenges, and facilitators. Folia Phoniatrica et Logopaedica. 2019; 72(2):131-42. [DOI:10.1159/000503131] [PMID]
- [19] Masso S, McLeod S, Baker E. Tutorial: Assessment and analysis of polysyllables in young children. Language, Speech, and Hearing Services in Schools. 2018; 49(1):42-58. [DOI:10.1044/2017_LSHSS-16-0047] [PMID]
- [20] Shakibayi MI, Zarifian T, Zanjari N. Speech characteristics of childhood apraxia of speech: A survey research. International Journal of Pediatric Otorhinolaryngology. 2019; 126:109609. [DOI:10.1016/j.ijporl.2019.109609] [PMID]
- [21] Iuzzini-Seigel J, Hogan TP, Green JR. Speech Inconsistency in children with childhood apraxia of speech, language impairment, and speech delay: Depends on the stimuli. Journal of Speech, Language, and Hearing Research: JSLHR. 2017; 60(5):1194-210. [DOI:10.1044/2016_JSLHR-S-15-0184] [PMID]
- [22] Murray E, Iuzzini-Seigel J. Efficacious treatment of children with childhood apraxia of speech according to the international classification of functioning, disability and health. Perspectives of the ASHA Special Interest Groups. 2017; 2(Part 2):61-76. [DOI:10.1044/persp2.SIG2.61]
- [23] Williams AL, McLeod S, McCauley RJ. Interventions for speech sound disorders in children. United States: Paul H. Brookes Pub.; 2010. https://books.google.com/ books/about/Interventions_for_Speech_Sound_Disorders. html?id=EZ0sQQAACAAJ
- [24] Gillon GT, Moriarty BC. Childhood apraxia of speech: Children at risk for persistent reading and spelling disorder. Seminars in Speech and Language. 2007; 28(1):48-57. [DOI:10.1055/s-2007-967929] [PMID]
- [25] Lefebvre P, Gaines R, Staniforth LA, Chiasson V. Emergent literacy skills in English-speaking preschoolers with suspected childhood apraxia of speech: A pilot study. Canadian Journal of Speech-Language Pathology and Audiology. 2017; 41(Special Issue):128-42. https://cjslpa.ca/files/2017_ CJSLPA_Vol_41/No_01/CJSLPA_Vol_41_No_1_2017_Lefebvre_et_al_128-142.pdf

This Page Intentionally Left Blank